Author’s response to reviews

Title: The no-go zone: a qualitative study of access to sexual and reproductive health services for sexual and gender minority adolescents in Southern Africa

Authors:
Alex Müller (alex.muller@uct.ac.za)
Sarah Spencer (scl.spencer@gmail.com)
Talia Meer (talia.meer@uct.ac.za)
Kristen Daskilewicz (ka.daskilewicz@uct.ac.za)

Version: 2 Date: 18 Jan 2018

Author’s response to reviews:

Dear editors,

As requested, we have placed two multilingual abstracts (one in French, one in Portuguese) after the official English language abstract in our manuscript.

Best regards

Alex Muller